brought to you by CORE



d'Economia de Barcelona

Document de treball de l'IEB 2010/36

THE EFFECT OF SUB-NATIONAL BORROWING CONTROL ON FISCAL SUSTAINABILITY: HOW TO REGULATE?

Violeta Vulovic

Fiscal Federalism

Document de treball de l'IEB

THE EFFECT OF SUB-NATIONAL BORROWING CONTROL ON FISCAL SUSTAINABILITY: HOW TO REGULATE?

Violeta Vulovic

The **IEB** research program in **Fiscal Federalism** aims at promoting research in the public finance issues that arise in decentralized countries. Special emphasis is put on applied research and on work that tries to shed light on policy-design issues. Research that is particularly policy-relevant from a Spanish perspective is given special consideration. Disseminating research findings to a broader audience is also an aim of the program. The program enjoys the support from the **IEB-Foundation** and the **IEB-UB** Chair in Fiscal Federalism funded by Fundación ICO, Instituto de Estudios Fiscales and Institut d'Estudis Autonòmics.

The **Barcelona Institute of Economics (IEB)** is a research centre at the University of Barcelona which specializes in the field of applied economics. Through the **IEB**-**Foundation**, several private institutions (Caixa Catalunya, Abertis, La Caixa, Gas Natural and Applus) support several research programs.

Postal Address: Institut d'Economia de Barcelona Facultat d'Economia i Empresa Universitat de Barcelona C/ Tinent Coronel Valenzuela, 1-11 (08034) Barcelona, Spain Tel.: + 34 93 403 46 46 Fax: + 34 93 403 98 32 <u>ieb@ub.edu</u> http://www.ieb.ub.edu

The IEB working papers represent ongoing research that is circulated to encourage discussion and has not undergone a peer review process. Any opinions expressed here are those of the author(s) and not those of IEB.

THE EFFECT OF SUB-NATIONAL BORROWING CONTROL ON FISCAL SUSTAINABILITY: HOW TO REGULATE?

Violeta Vulovic

ABSTRACT: This article examines effectiveness of sub-national borrowing control regimes in maintaining overall fiscal sustainability. The results suggest that regulating sub-national borrowing based on fiscal rules performs most efficiently in maintaining fiscal consolidation. Furthermore, sole reliance on financial markets seems to lead to faster end of fiscal consolidation episodes, which may be explained by not fully developed financial markets in many countries that dominantly apply this approach, Finally, strong central government control, as in case of administrative and cooperative regimes, in presence of high fiscal dependence on central government financing seem to increase the probability of ending consolidation episodes.

JEL Codes: H61, H63, H71, H72, H74, H77

Keywords: Sub-national borrowing, fiscal sustainability, panel data

Violeta Vulovic Georgia State University 517 Main Street Apt 2229 Atlanta GA 30324 Phone: +1 404 906 4359 E-mail: <u>vvulovic@gsu.edu</u>

I. Introduction

Decentralization of borrowing authority to sub-national government and fiscal sustainability at the national level are two issues in permanent tension in public financial management. On one side of the argument, it is desirable to give sub-national authorities room for raising their own financial resources in order to finance capital investment for the provision of goods and services to their communities. On the other side of the argument, the lack of institutional capacity, history of sub-national government defaults in other decentralized systems, and the potential lack of effective controls at the least give central governments substantial arguments to restrict sub-national government autonomy. Thus, the challenge is whether is possible to simultaneously achieve the goals of providing borrowing autonomy and maintain fiscal discipline preventing the insolvency of sub-national governments.

Furthermore, sub-national governments have less incentive than the central governments to be concerned with macroeconomic impact of their policies because they do no bear the full cost of their actions ("moral hazard"). Therefore, to sub-national governments, macroeconomic stability has public goods characteristics. Some authors contend that fiscal decentralization can enhance macroeconomic stability (Fukasaku and De Mello, 1998) while others assert there are significant costs associated with insuring macroeconomic stability through decentralization and that achieving this goal requires thoroughly disciplined sub-national borrowing (Ter-Minassian, 1997). The empirical literature on this issue is inconclusive. It is therefore important to investigate the effect of fiscal decentralization on macroeconomic stability in the presence of borrowing regulation.

Fiscal decentralization is increasingly seen as a tool to promote economic efficiency. Oates (1993) explains, "the provision of local outputs that are differentiated according to local tastes and circumstances results in higher levels of social welfare than centrally determined and more uniform levels of outputs across all jurisdictions." (p. 240). Two basic mechanisms are involved here. The first mechanism relates to Hayek's (1945) knowledge problem, which states that wide dispersion of knowledge causes central planning to fail.¹ Decentralized authorities, on the other hand, are in much better position to be responsive to variations in local demand. The second mechanism is related to the idea government as a monopolist. As Brennan and Buchanan (1980) suggest, "the potential for fiscal exploitation varies inversely with the number of competing governmental units in the inclusive territory." (p. 180). Therefore, greater competition between governments can limit their ability to extract monopoly rents, enhancing economic efficiency and economic growth.

¹ "The economic problem of society is . . . a problem of the utilization of knowledge not given to anyone in its totality. . . . [This] is at least one of the main problems of economic policy - or of designing an efficient economic system. . . . This is not a dispute about whether planning is to be done or not. It is a dispute as to whether planning if to be done centrally, by one authority for the whole economic system, or to be divided among many individuals" (pp. 519–521).

[&]quot;If we can agree that the economic problem of society is mainly one of rapid adaptation to changes in the particular circumstances of time and place, it would seem to follow that the ultimate decisions must be left to the people who are familiar with these circumstances, who know directly of the relevant changes and of the resources immediately available to meet them. We cannot expect that this problem will be solved by first communicating all this knowledge to a central board which, after integrating *all* knowledge, issues its orders. We must solve it by some form of decentralization. . . . We need decentralization because only thus can we ensure that the knowledge of the particular circumstances of time and place will be promptly used" (p. 524).

Due to widespread decentralization of spending responsibilities, increasing revenue power and borrowing capacity of sub-national governments, sub-national borrowing has become an increasingly important source of sub-national finance. Proponents of sub-national borrowing emphasize four benefits: (i) expansion of sub-national fiscal space for infrastructure financing; (ii) efficient and inter-generational equitable outcomes from infrastructure financing through borrowing; (iii) increased fiscal transparency of sub-national governments; and (iv) deepening of financial markets. However, while there is considerable consensus on those potential benefits, there is also wide agreement that without an effective regulatory framework sub-national borrowing may lead to fiscal and debt crises and significantly contribute to an unstable macroeconomic environment.

For certain, appropriate regulatory frameworks where borrowing or deficit financing is only allowed to finance capital investments (the so called "golden rule") accompanied by limits on the level of debt and debt servicing capacity can reduce the chances of default and debt crises. However, other institutional factors must be present. In particular, sub sub-national governments must have access to significant tax bases, because otherwise, even if borrowing is put into productive use, it may still cause fiscal crises. Dependence on inter governmental transfers might lead to unsustainable borrowing since high levels of transfer dependence often undermine the credibility of the central government's commitment not to bail out troubled sub-national governments. By a similar logic, when sub-national governments are funded primarily by the taxes they raise and collect themselves, the central government can commit more easily to a no bail out policy, thus giving creditors and voters stronger signals and incentives to "punish" sub-national officials for excessive spending and borrowing.

Despite the importance of these issues, little systematic empirical work has been done so far on the role of sub-national borrowing and borrowing control on fiscal sustainability. Therefore, the question this study tries to answer is whether controlled sub-national performs better or worse comparing to prohibition of borrowing in maintaining fiscal sustainability, and if yes, which setting performs superiorly and under which circumstances.

These issues are of particular importance because rapidly rising sub-national debt has played a crucial role in recent financial crisis in several countries and those experiences hold important lessons for other countries undergoing fiscal decentralization. The few studies that evaluate the effect of sub-national borrowing on fiscal performance either use only debt level or some aggregate measure of borrowing autonomy that does not take into account different types of regulations, monitoring and enforcement. From our perspective, qualitative indicators of fiscal decentralization, such as taxing powers or autonomy of subnational governments to decide on tax base and tax rates, different forms of borrowing powers and regulation and enforcement, must be considered to avoid obtaining biased and misleading empirical results on the effect of sub-national borrowing on fiscal performance.

This study employs a panel data set of 60² countries, between 1990 and 2008. Variables of primary interest are qualitative indicators on sub-national borrowing control regimes which countries in the sample employ during the observed period. The information on these qualitative indicators is collected by the author from various sources³. Because sub-national borrowing control regimes are not mutually excludable, and countries usually implement more than one set of controls, the qualitative indicator for each country in the sample represents the regime which is dominantly applied. Fiscal sustainability is defined based on

² See Appendix for the list of countries.

³ For the list of sources by country, see Table A4 in the Appendix.

annual change in general government budget balance and budget is considered as sustainable if the annual change is equal or greater that some predetermined threshold. This study uses zero percent change as baseline threshold and, for the purpose of robustness analysis, applies alternative thresholds of 0.5, 0.75 and 1 percent. The applied methodology is survival analysis which is appropriate for evaluating the effect of each borrowing control regimes on duration of fiscal sustainability.

Obtained results suggest fiscal rules to perform dominantly in sustaining fiscal consolidation. Second, results suggest that market discipline performs the least favorably comparing to other three approaches and prohibited sub-national borrowing, which corresponds to the fact that in many countries the requirements for market discipline as a successful instrument for sub-national borrowing control are not met. Finally, administrative and cooperative regimes in presence of high fiscal dependence on central government financing seem to increase the probability of ending a consolidation episode. This result is not surprising given that these two regimes refer to significantly high degree of central government control, which may increase the risk of moral hazard. Strong central government control may give encouraging signs to the sub-national governments to over borrow and to expect being bailed out by the central government. This deteriorates the general government budget directly, through unplanned bailout from the central government, and indirectly, through spillover effect on other sub-national governments who are as well highly fiscally dependent on central government financing. Finally, the robustness analysis suggests that the rule based and administrative controls over sub-national borrowing consistently perform dominantly comparing to other regimes.

The results obtained in this study imply following important policy recommendations. First, sub-national government borrowing does not have to endanger overall fiscal sustainability if the borrowing control framework is well designed. Second, reducing fiscal dependence on central government financing and, in return, giving more tax autonomy to sub-national government, reduces the risk of moral hazard and improves the effectiveness of borrowing control in maintaining fiscal balance at the sustainable level.

The rest of this paper is organized as follows. Section II provides brief literature review on the effect of fiscal decentralization and sub-national borrowing on macroeconomic stability. Section III describes institutional background of four broad approaches that have been used to regulate or control the operation of sub-national credit markets. Section IV provides details on applied methodology. Section V explains results obtained using baseline threshold and from robustness analysis. Finally, section VI concludes the study.

II. Literature Review

The first theoretical examination of fiscal decentralization was undertaken by Tiebout (1956), Musgrave (1959), and Oates (1972). Fiscal decentralization is widely advocated with an argument that it improves economic performance by increasing economic efficiency in the provision of public sector services. Some, however, argue that fiscal decentralization can impact economic performance when sub-national governments have uncontrolled expenditures, which adversely affects national fiscal policy and macroeconomic stability.

There is a large and wide-ranging literature investigating economic benefits of fiscal decentralization that are claimed by its advocates. Majority of these studies focuses on the effect of either revenue or

expenditure decentralization on economic performance, with just few exemptions considering the relationship between borrowing autonomy and fiscal sustainability (De Mello, 2001; Martinez-Vazquez and Boex, 2001; Rodden, 2002; Wibbels and Rodden, 2007; Martell, 2008).

Decentralization and Macroeconomic Stability

Large part of recent literature focuses on the macroeconomic problems that arise with devolving greater responsibilities to the sub-national governments. Empirical results obtained by Prud'homme (1994), Hunter and Shah (1996) and Ter-Minassian (1997) suggest that decentralization results in sub-national fiscal indiscipline and aggravated fiscal problems at the central level. De Mello (2000) finds that in developing countries expenditure devolution tends to worsen central government balance. Fornasari and Steven (2000) also find that increases in sub-national spending leads to increases in national spending and deficits. However, Stein (1999) shows that decentralization is not associated with higher deficits in Latin America. Similarly, Shome (2002) finds that decentralization is associated with lower fiscal deficits both at the state and central government levels in India. ⁴

However, some authors emphasize that decentralization can also cause macroeconomic problems when key institutional and financial pillars are absent from the federal framework. Coordination difficulties between the various layers of government can challenge macroeconomic sustainability even in the least decentralized of systems. Various studies (Tanzi, 2000; Dabla-Norris and Wade, 2002) find that incentives for responsible fiscal behavior and hard-budget constraints are undermined when the federal framework is characterized by a (i) lack of local autonomy over expenditure and revenue decisions, and a high degree of dependence on transfers; (ii) lack of constraints on sub-national indebtedness; (iii) lack of clarity in the respective roles of each tier of government; and (iv) weak budget institutions. In his study, Rodden (2002) shows that decentralization is associated with large and persistent general government deficits when sub-national governments are simultaneously dependent on transfers and are free to borrow.

How to regulate sub-national borrowing?

There are different ways in which the national government can contribute to prudent borrowing. Which option to take is a much-debated issue (Peterson, 2000). Literature on sub-national borrowing emphasizes the ability of higher levels of government to provide an implicit guarantee on sub-national government debt as one of main problems with borrowing at the sub-national level. This is a classical moral hazard situation, whereby borrowers are likely to over borrow and creditors are likely to over lend in response to this unwritten official insurance. The question is whether such a risk can be successfully controlled by some kind of rule, or if the credit market can do the job on its own. A fundamental decision that a national government takes final responsibility in dealing with a financial crisis of sub-national entities. If a sub-national entity is unable to re-pay debt, the national government would step in and bail out the failed creditor. Each country needs to decide how to deal with the above-mentioned challenges when establishing a legal framework. National governments have adopted different responses to the challenges of decentralized decision-making. A key question is how a country chooses to control sub-national borrowing in order to avoid the risks associated with it.

Based on a sample that includes Indonesia, Mexico, the Philippines, Poland, and South Africa, Martell and Guess (2006) consider the legal framework to be of paramount importance and suggest establishing such a

⁴ Except when transfers are excluded. The inability of states to fund their own-expenditure without central government transfers results in higher state-level deficits.

framework first in a sequence of reforms, followed by a viable supply-side and a creditworthy demand-side of the borrowing market. A regulatory framework should at least deal with three challenges:

- First, a regulatory framework must not prohibit sub-national borrowing. Rather, the legal framework should be such that sub-national entities are allowed to engage on their own with financial markets to finance their projects. In that sense, the regulatory framework enables demand by devolution of borrowing power to sub-national entities. It is furthermore important to stipulate which levels of government may borrow. This helps regulating the interaction of different levels of government.
- Second, a regulatory framework should provide predictability, clarity and confidence in sub-national borrowing. Only a clearly stated legal framework can encourage participants, ranging from investors to municipal officers to engage with sub-national borrowing. To play this overarching role, the design of the framework is crucial: The framework needs to be well formulated, comprehensible and consistent. It also has to cover all necessary aspects.
- Third, a good regulatory framework can reduce the risk of imprudent borrowing by preventing overborrowing and by providing instructions on how to deal with financial crises. Over-borrowing at the sub-national level and instability at the macroeconomic level in form of financial deficits or inflation are less likely to happen when there is a good system of regulations.

Empirical support for the hypothesis that institutional constraints limit government spending is not fully conclusive. While rare cross country evidence shows that the effectiveness of institutional constraints heavily depends on the type of control being imposed and a number of idiosyncrasies of the country in question (Plekhanov and Singh, 2006), there is mixed evidence from country-level studies conducted in the United States and Europe. For other sources of debt finance in the United States, Abrams and Dougan (1986) conclude that restrictions on borrowing and spending are not significant in explaining state budget outcomes. The French case appears to suggest that macroeconomic policy measures emanating from the central government do affect local government borrowing decisions (Derycke and Gilbert, 1985). Less conclusive results are obtained by Kenyon (1991) on the effects of caps on federal and local tax-exempt bond issues in the United States. While caps were shown to be effective in reducing the volume of borrowing, they do not appear to have a significant impact on whether sub-national governments substitute tax-exempt bonds for other sources of borrowing.

Further confirmation that institutional restrictions do matter can be found in Alt and Lowry (1994), who examine the effectiveness of state-level fiscal control in the United States. Their empirical results suggest that divided party governance matters when it comes to responding to exogenous shocks. Interestingly, they also find significant differences between Democrats and Republicans when it comes to fiscally relevant decisions. Although more cautious in arriving at general conclusions, Poterba (1995) provides further confirmation of the role of fiscal rules in the United States. The opposite result can be found in a study of Spain by Cabases et al. (2007). This provides evidence that municipalities are sensitive to institutional restrictions on their decisions to borrow. On the one hand, borrowing appears to be used mainly for investment as established by the law. On the other hand, restrictions on short-term and emergency borrowing based upon a maximum percentage of the previous year's revenues significantly affect levels of current-year indebtedness. The type of municipality and the level of local co-funding also matter as far as municipal borrowing is concerned.

III. Institutional Background: Sub-national borrowing control regimes

In order to have a better understanding of the impact of sub-national borrowing, it is desirable to review the different institutional settings that have been used to regulate or control the operation of sub-national credit markets. In this section we review four main, not mutually excludable, settings that can be found in the international practice, namely: (i) market discipline; (ii) rule-based controls; (iii) administrative controls; and (iv) cooperative approach (Ter-Minassian and Craig, 1997).

III.1 Market Discipline

Some countries rely exclusively on capital markets to restrain sub-national borrowing. Market discipline means that financial markets are capable to send appropriate signals to prevent a borrower from entering the "unsustainable area." There are, however, certain preconditions that need to be satisfied for financial private markets to be an effective control instrument for sub-national borrowing. These include (Lane, 1993): (i) capital markets must be free and open; (ii) potential lenders must have available information about the borrower's outstanding debt and repayment capacity; (iii) there should be no chance or possibility of bailout of lenders by the central government; and (iv) borrowers must have the ability to respond with adequate policies to the signals sent by the market.

In this sort of setting, sub-national governments have generally direct access to financial markets to meet their borrowing requirements. Also they independently decide how much and from whom to borrow, and on what to spend the borrowed money. Relying on market discipline in controlling the sub-national borrowing has worked in countries with high standards of transparency and governance at all government levels, and with no significant history of bailouts. In most emerging markets and developing countries, however, as well as in some developed countries, one or more the requirements for effective market discipline on sub-national borrowing are missing. In particular, information on sub-national finance is often not accurate and comprehensive enough, sub-national governments often have access to borrowing funds under "privileged" terms (e.g. municipal development banks or sub-national enterprises), and many have significant bailout histories (through fiscal gap filling transfers or through debt restructuring). Moreover, existence of implicit guarantee by the central government prevents market signals to work effectively.

Examples of market disciple in controlling the sub-national borrowing can be found, among some others, in provinces in Canada, the U.S. states and Swedish municipalities. However, as mentioned above, in many parts of the world, the capital markets at the local level are inadequately developed to be able to efficiently discipline sub-national governments. In such circumstances, credit rating agencies at the sub-national level are becoming increasingly important to evaluate the performance of intergovernmental system. In this same context, some sub-national governments have adopted fiscal responsibility rules (that are self-imposed) trying to improve their credit rating in the market. Examples for these trends are present in Canada, Switzerland, and the United States. Some countries in Latin America, such as Argentina, Brazil, Colombia and Peru, recently have sought to follow this approach, at least partially, with the introduction of a Fiscal Responsibility Laws (Webb, 2004).

III.2 Rule Based Approach

Rule-based controls consist of fiscal rules imposed by the central government and specified in the constitution or in organic laws. Such rules introduce a constraint on fiscal choices by sub-national governments to guarantee that fiscal outcomes will remain predictable and robust regardless of the government in charge. Rules may take different forms: ceilings on debt or total borrowing, deficit targets,

maximum expenditure rules, the "golden rule" (borrowing proceeds must be spent exclusively on capital projects), or rules related to debt payment capacity.

Debt ceilings are in general simple and easy to monitor. A deficit target has the advantage of simplicity as well, and of being easily understood by the wider public, but it may be unsuccessful in preventing excessive debt accumulation because of off-budget items. The most frequent deficit target rules are those targeting the overall budget deficit (for example, Austria, Belgium, Spain, and most U.S. states) or operating deficit (for example, Norway). However, deficit target rules can be met with higher revenues and expenditures as well, which may have macroeconomic implications. Expenditure rules set the limits on expenditure level, and are conceptually simple, easy to monitor, and can be most directly controlled. The golden rule, limiting sub-national governments' borrowing to investment purposes only, mostly satisfies the intergenerational equity justification for borrowing. However, borrowing for infrastructure does not guarantee by itself macroeconomic and debt stability. Typically, infrastructure investments are required to provide "adequate" economic and social rates of return to be desirable or be approved. Many countries currently implement some form of the golden rule (for example, the United Kingdom, Germany, Spain, and most states in the U.S.) Rules related to debt repayment capacity attempt to simulate the workings of the market discipline approach by relating the limits on the indebtedness to expected debt service on the debt.

Fiscal rules have the advantage of being generally transparent, more effective in addressing long-term sustainability and intergenerational equity, depending on the accuracy of financial indicators, and relatively easy to monitor. Their effectiveness, however, depends on their specificity, comprehensiveness of coverage, and, most importantly, the degree of political commitment to their observance and enforcement. They as well can be counterproductive if poorly designed, that is, if in particular, there is no clear specification of appropriate escape clauses and of credible sanctions for noncompliance. It is needless to say that the availability of information to timely access compliance is crucial to the fiscal rules' effectiveness. Most countries using a rule-based approach use a variety of rules, sometimes redundant.

III.3 Administrative Approach

This approach gives the central government direct control over sub-national borrowing. Administrative controls may take different forms that vary in comprehensiveness and degree of detail. Examples are setting an annual (or even more frequently) limits on the overall sub-national government debt; prohibition of external (foreign) borrowing; review and approval of individual borrowing operations (including approval of the terms and conditions); or the centralization of all government borrowing with on-lending to sub-national governments. Administrative controls have tended to focus more on overall levels of borrowing and debt, rather than on authorization of individual borrowing operations.

Countries like Greece, Ireland, Mexico, or the United Kingdom practice administrative controls. In Mexico, the states and municipalities, including their decentralized agencies and public enterprises, can only borrow domestically to finance investment outlays up to the ceilings set by their respective legislatures. Unlike several other countries in Latin America, Mexico does not have a Fiscal Responsibility Law even under consideration. It uses financial sector regulations instead to motivate state-level prudence. In the United Kingdom, a local authority may not, without the consent of the Treasury, borrow from a lender from abroad or other than in sterling. In Spain, for example, foreign debt and bond issuances by sub-national governments are subject to the approval of the ministry of finance.

Administrative controls, especially when central authorization of individual sub-national borrowing is involved, carry a significant moral hazard, resulting from the fact that central government may find it difficult

to refuse to financially support the lower levels of government in the case of impending defaults. On the other hand, the administrative approach has several advantages. First, the central government can control both the macroeconomic and external debt policy. Second, central government's control may increase subnational borrower's credibility, given that foreign lenders often require a central government guarantee, and it is likely result in better terms and conditions received in foreign financial markets. In overall, the effectiveness of administrative approach in controlling sub-national borrowing and promoting their fiscal discipline depends crucially on the de-politicization of the central government's decisions, and on availability of information needed to shape these decisions and check their enforcement. ⁵

III.4 Cooperative Approach

Under this approach, sub-national borrowing controls are designed through a negotiation process between the federal/central and the lower levels of government. Sub-national governments are actively involved in reaching an agreement on overall general government deficit targets, on main revenue and expenditure items, as well as on the limits on financing of individual sub-national jurisdictions. This approach is in practice in some European countries and in Australia.

In Austria, for example, the "Consultation mechanism" between different levels of government and the "Stability pact" were implemented in 1999 (Thoni, Garbislander, and Haas, 2002) to ensure lowering and maintaining the overall deficit below 3 percent. Similar arrangements exist in Spain (Lopez-Laborda et al. ,200??). In Belgium, the sub national borrowing is supervised by a High Finance Council (HFC), which comprises members nominated by the federal, regional, and community levels, and the Belgian National Bank. In Australia, a fiscal institution called the Loan Council coordinates the fiscal policies and borrowing decisions of Australian states. Brazil and Argentina as well promote cooperation between levels of government.⁶

The cooperative approach combines many individual advantages of the other three approaches which is both its main strength and its main weakness. A clear advantage lies in promoting dialogue and exchange of information across various government levels, as well as in raising awareness of the macroeconomic implications of their budgetary choices. However, the preconditions for its success include the absence of severe fiscal stress; relative homogeneity of the sub-national units; a tradition of cooperation in intergovernmental relations; relatively strong bargaining position of the central government to be able to effectively guide the intergovernmental negotiations, which in many emerging markets' conditions may not be the case; and availability of reliable and timely information to access compliance with agreed borrowing limits. Finally, weakness of the cooperative approach is that when it is poorly implemented it reproduces the flaws of other approaches, instead of their advantages (Ahmad, Albino, and Singh, 2005).

IV. Methodology and Variables

IV.1 Methodology

Two alternative approaches can be undertaken to assess the determinants of fiscal sustainability using survival analysis, namely gradient approach and level approach (Adam and Bevan, 2003). Under the gradient approach, fiscal adjustment ends when a country fails to keep reducing the deficit by a certain

⁵ Including information on off-budget activities of the sub-national government, and on incurred arrears.

⁶ Brazil through debt restructuring agreements with states and municipalities, and Argentina through bilateral pacts between the nation and the provinces.

threshold amount each year. Under the level approach, the end of fiscal consolidation episode is reached when the deficit goes above a certain deficit threshold. In this study a modified gradient approach is chosen with 0 percent threshold. A fiscal adjustment is considered as continuing if budget balance as a share of total expenditure changes by more than 0 percentage points per year.

This study employs annual panel data between 1990 and 2008 for 60⁷ developed, developing and countries in transition. To define the dependent variable, this study uses general government⁸ primary⁹ balance (Budget Balance), as a share of gross domestic product (GDP). The reason for choosing general government rather than primary budget balance at the sub-national level is the following: When the central government faces the sub-national fiscal imbalances, it can react in the following three ways. First option for the central government is to cover the sub-national imbalances. Second option is to design the tax and/or transfer system through which the sub-national government would receive larger portion of the overall revenues collected. Finally, the third option is that the central government ignores the sub-national fiscal imbalances is likely to deteriorate.

Based on these data, a dummy variable called "Failure" is generated, which takes value zero when the annual variation of Budget Balance is greater or equal zero (years of fiscal consolidation), and takes value one when the annual variation is lower than zero (years of fiscal expansion). Using the dates in which failure event occurs, a new variable called "Duration" is built, which counts the intervening years between consecutive failures (the time span that fiscal consolidation lasts). In our sample, the minimum number of years that a consolidation lasts is one year, and the maximum is fifteen years.

Table1 presents the structure of data on Failure and Duration. The total number of observations is 847 and the average duration of fiscal consolidations is around 2 years, the maximum duration being 10 years. The number of registered failures is 330, and the average probability of ending a fiscal consolidation is 39 percent. Figure1 shows the duration of fiscal consolidations in the period 1990-2008, where 44 percent of fiscal consolidations lasts one year, 25 percent two years, 13.7 percent lasts three years, and 17.3 percent lasts four and more years.

⁷ List of countries provided in the Appendix.

⁸ The general government sector consists of entities that fulfill the functions of government as their primary activity and can be divided into central, state, and local government subsectors, depending on a country. In the Government Finance System (GFS), statistics for the general government sector and each of its subsectors are presented on a consolidated basis, to avoid the double counting of transactions. Consolidation involves the elimination of all transactions "that occur among the units being consolidated. In other words, a transaction of one unit is paired with the same transaction as recorded for the second unit and both transactions are eliminated ... For example ... consolidated interest revenue and expense exclude the interest paid by the debtor general government unit to the creditor. Similarly, sales of goods and services between consolidated units are also eliminated." (Internatinal Monetary Fund, 2001)(Internatinal Monetary Fund, 2001: 33)

⁹ Revenues – Expenditures + Interest Payments

	Failure	Duration
Mean	0.390	2.267
Standard Deviation	0.488	1.164
Variance	0.238	2.770
Skewness	0.453	1.819
Kurtosis	1.205	6.742
Number of failures	330	
Observations	847	



A. Non-parametric Estimation

The time dependency of duration. This section proceeds very briefly with the non-parametric analysis, which tries to investigate whether fiscal consolidation is positively or negatively dependent on their accumulated duration. This is typically done by estimating the two following functions:

- The survivor function gives the probability that the duration of the fiscal consolidation (7)¹⁰ is greater or equal to t, and is defined as
 S(t) = Pr(T ≥ t) = 1 F(t) (1)
- 2. The hazard function gives, for each duration, the probability of ending a consolidation episode, conditioned on the duration of the consolidation through that moment, and is defined as

 $^{^{10}}$ T is the discrete random variable that measures the time that passes between the beginning of a fiscal consolidation and its transition to a non-consolidation period.

$$h(t) = \Pr\left(T = \frac{t}{T} \ge t\right) \tag{2}$$

Figure 2 shows the estimated survivor function, where we can see that about 30 percent of episodes lasts 3 years or more.



The estimated hazard function, in Figure 3, gives evidence of positive dependence of fiscal consolidations on their accumulated duration.



B. Parametric Estimation

Non-parametric analysis does not allow one to analyze other factors other than accumulated duration that may explain the probability of ending fiscal consolidations. To address this issue, we estimate a Model of Proportional Hazard (PH), which assumes that the hazard function can be split as follows:

$h(t,X) = h_0(t) \cdot g(X)$

where $h_0(x)$ is the baseline hazard that captures the dependency of data to duration, while g(x) is a function of individual variables. This function of explanatory variables is a negative function, usually defined as $g(x) = \exp(x'\beta)$, so model has the following form:

$h(t,X) = h_0(t) \cdot \exp(X'\beta)$

This model can be estimated initially without imposing any specific functional form on the baseline hazard function, following the Cox Model. An alternative estimation can be done by imposing one specific parametric form to the function $h_0(c)$. In this case, the models most commonly used are the Weibull Model and the Exponential Model. In the Weibull Model, $h_0(c) = pt^{p-1}$, where p is a parameter that has to be estimated. When p = 1, the Weibull Model is equal to the Exponential Model, where there exists no dependency on duration. On the other hand, when the parameter p > 1, there exists a positive dependency on duration, and a negative dependency when p < 1. Therefore, by estimating p it is possible to test the hypotheses of duration dependency of fiscal consolidations.

Apparently, the case examined in this study is a case of multiple failure-time data where more than one failure occurs for the same subject (country), causing failure-times to be correlated within cluster (country), violating the independence of failure-times assumption required in traditional survival analysis. If more than one spell is observed for a country, it is realistic to assume that these spells are not independent. Thus, likelihood function based on model (4) is misspecified for multiple spells since it does not account for intra-country correlation of the spells observed on the same country.

Following (Lin & Wei, 1989), it is necessary and sufficient to modify only the variance-covariance matrix of the estimators since the correlated durations affect the variance while the model parameters can be estimated consistently without accounting for this correlation. This implies that parameters of the model can be estimated by treating spells as independent, and then obtained variance and covariance estimates can be modified to account for the dependences. More precisely, given that estimated variance-covariance matrix obtained as the inverse of the information matrix does not take into account the additional correlation in the data, Lin and Wei (1989) propose a modification of the following form:

$V = I^{-1}(\beta)G'(\beta)G(\beta)I^{-1}(\beta)$

(5)

where $G(\mathcal{G})$ is a $m \times p$ matrix of the group efficient score residuals (m is the number of clusters $(G_1 1, G_1 2, ..., G_1 m)$, while p is the number of time-dependent covariates).

IV.2 Variables

(3)

(4)

Variables of interest are alternative borrowing control regimes, namely administrative, rule based, cooperative regime and market discipline, with base category including countries in which borrowing is prohibited at the sub-national level.

Control Variables

Vector of control variables includes a set of institutional, economic, political, and demographic variables that are expected to be related to different demands for expenditure financing and, therefore, different need for borrowing. Hence, these demands put a pressure on sub-national fiscal performance and affect the length of fiscal consolidation.

The control variables are:

1. *Number of previous failures*: this variable controls for the accumulated number of ends of fiscal consolidations that have taken place in each country before the current consolidation. It is, ceteris paribus, expected that the larger this number, the higher is the probability an episode of fiscal consolidation would end.

2. *Initial budget balance*: this variable takes into account the fact that initial fiscal consolidations influence policymakers in deciding how much adjustment is needed to stabilize the public finances. This controls for the fact that countries with high budget balance may not feel compelled to continue with fiscal adjustment, as the balance may already be close to sustainable level. Therefore, this variable is expected to increase the probability of ending the consolidation episode.

3. *The size of the fiscal adjustment*: this variable is measured as a cumulative change in the budget balance during the fiscal consolidation episode. The larger the size of the consolidation, the longer the episode is hypothesized to last, because a larger adjustment tends to signal the willingness of the authorities to achieve fiscal sustainability.

4. *Central government budget balance:* Rodden (2002) argues that sub-national government fiscal performance may be positively correlated with the central government long-term fiscal performance. Plekhanov and Singh (2006) provide the following three reasons for this relationship. First, the average central government fiscal balance is a proxy for the society's preference toward the fiscal sustainability. Second, the average central government fiscal balance captures possible business cycle effects, especially for countries having few observations on sub-national government fiscal balance. Third, central government fiscal balance partly absorbs the effects of fiscal crises that affect the fiscal performance. Therefore, longer consolidation episodes are expected in sub-national and general government budgets in countries with well behaved central governments.

5. *Vertical Fiscal Imbalance*: this variable is measured as a percentage of sub-national received grants of their total revenue and represents the sub-national government dependence on the central government. The literature provides the evidence that higher vertical fiscal imbalance is positively correlated with sub-national fiscal indiscipline and high spending, hence with shorter consolidation episodes.

6. *Fiscal decentralization*: fiscal decentralization is measured as a share of sub-national own revenues in total general government revenues Prud'homme (1994) argues that one of the dangers of decentralization is that it makes macroeconomic stabilization programs more difficult to implement because sub-national government fiscal policies can run counter to national policies. It can, therefore, lead to worse

fiscal outcomes. Singh and Plekhanov (2006) indicate that it may also reflect the central government's attempt to shift part of the fiscal burden onto sub-national governments. On the other side, (Shah, 2005) finds that fiscal decentralization is associated with improved fiscal performance and better functioning of internal common market. Hence, the effect of fiscal decentralization on duration of consolidation episode is ambiguous.

7. *Sub-national tax autonomy*: variable tax autonomy takes value one if sub-national government has an authority to set tax rate and/or change the tax base. It is expected that sub-national governments with more tax autonomy are better able to optimize their revenues to their expenditure needs and avoid jeopardizing their fiscal balance. This variable is therefore expected to be associated with longer consolidation periods.

8. *Borrowing for only investments purposes* as opposed to allowing borrowing for deficit financing is expected to have positive effect on primary fiscal balance trough more than one channel. First, prohibiting borrowing for financing fiscal deficit positively affects sub-national fiscal responsibility. Second, borrowing for capital investments has potentially positive effect on increasing the sub-national revenue base in the long run and, through it, potential higher revenue collection.

9. *Corruption* is measured by the survey-based perception index which takes value between zero and six, with higher index meaning lower corruption. Perceived corruption is assumed to be associated with weak government institutions and, therefore, lower fiscal discipline and higher probability of an end of consolidation episode.

10. *Government stability* is an assessment both of the government's ability to carry out its declared programs, and its ability to stay in office. The risk rating assigned is the sum of three subcomponents (government unity, legislative strength, popular support), each with a maximum score of four points and a minimum score of 0 points. A score of 4 points equates to Very Low Risk and a score of 0 points to Very High Risk. More stable governments are expected to more likely impose a harder budget constraint on all levels of government and may improve fiscal outcomes.

11. *GDP per capita growth:* this variable is meant to account for better fiscal performance of developed countries.

12. *Population density:* Increasing population density implies a higher cost of the publicly provided good due to congestion (Fenge & Meier, 2002) However, for some public goods, such as sewer, the cost can fall with increasing population density (Haug, 2004). This variable as well serves as a proxy for sub-national administrative capacity and ability to successfully administer sub-national borrowing. Lower population density may indicate existence of smaller size of sub-national units that usually lack of staff. Having all mentioned in mind, as in case of fiscal decentralization, the effect of population density on fiscal performance is ambiguos.

13. Finally, dummy for *European Union* member countries accounts for the "Maastricht effect" that imposes restrictions on fiscal behavior, both sub-national and national, and is expected to be associated with longer consolidation periods and lower probability of ending a consolidation episode.

V. Results

Table 2 presents parameter estimates for the three alternative previously discussed hazard function specifications. Country-dummies are included in all regressions to account for unobserved heterogeneity coming from individual country effects. The *p* parameter in the Weibull estimation is positive and greater than 1, indicating increasing hazard function over time.

Log likelihood ratio, Akaike Information Criteria (AIC) and Bayesian information criterion (BIC) is used to discriminate among three parametric models. Although the best fitting model is the one with the largest log likelihood, the preferred model is the one with the smallest AIC value. As Table 2 shows, the Weibull estimation is the parametric model that at the same time best fits our data (has the largest log likelihood) and is the most preferred (has the smallest both AIC and BIC value).

As results suggest, most of the control variables that are statistically significant have expected effect on probability of ending consolidation episode. Estimated coefficients on variables of our primary interest, borrowing control regimes, suggest that, comparing with an option of prohibiting borrowing at the subnational level, countries that primarily rely on rule based, cooperative and administrative regime are more effective in maintaining fiscal consolidation, rule based regime being the most successful (about 77 percent smaller hazard than prohibiting sub-national level perform worse in achieving fiscal sustainability than prohibit sub-national borrowing. This result may be explained by the requirements for successful market discipline that, as discussed above, in many countries, especially developing, are not fully met. However, our result suggest that the difference in the performance between prohibited and sub-national borrowing regulated by the marker is neither statistically nor economically significant.

On the other hand, results suggest that in presence of high sub-national revenue dependence on intergovernmental transfers from the central government, countries relying on cooperative and administrative controls have significantly higher probability of ending fiscal consolidation episode that countries prohibiting borrowing at the sub-national level. This result as well corresponds to the characteristics of these two regimes that both represent the highest level of central government control of sub-national borrowing, which in presence of high vertical fiscal imbalance may as well include larger probability of soft budget constraint. In this case, overall government fiscal balance suffers through two channels. First, existence of soft budget constraint results in lower sub-national fiscal responsibility. Second, bailing out the sub-national government deteriorates the central government budget and, through a spillover effect, budgets of other sub-national governments in the country that are as well highly fiscally dependent on intergovernmental transfers. Previously explanation of characteristics of the cooperative regime, that combines many individual advantages of the other three approaches, is in line with this result.

VI. Robustness Analysis

To check for robustness of the results, the parametric estimation from the previous section is replicated by using three alternative "Stronger" definitions of fiscal consolidation, namely, if an annual change in general government primary budget balance is less than 0.5, 0.75 and 1 percent. It can be said that the 0 percent threshold is the minimum threshold that one can impose to differentiate fiscal consolidation years from fiscal expansion ones. The 1 percent threshold is the most common in the literature on fiscal adjustments,

because it discriminates in favor of strong consolidation experiences, where the political commitment to reduce the public deficit is strong and cannot be attributed to unintended outcomes.

	(1)	(2)	(3)
	Weibull	Exponential	Сох
Number of previous failures	1.146***	1.075***	1.096***
·	(0.038)	(0.018)	(0.024)
Initial budget balance	0.980	0.991	0.988
5	(0.028)	(0.010)	(0.017)
Size of the adjustment	0.007**	0.042***	0.029***
,	(0.017)	(0.044)	(0.039)
Central Government Primary Balance	0.832	0.931	0.836
5	(1.893)	(0.826)	(1.108)
Administrative regime	0.306***	0.621**	0.493***
3	(0.136)	(0.125)	(0.132)
Cooperative regime	0.343**	0.658*	0.552**
	(0.166)	(0.153)	(0.165)
Rule based regime	0.230**	0.523**	0.422**
5	(0.135)	(0.139)	(0.147)
Market discipline	1.048	1.012	1.006
·	(1.536)	(0.701)	(0.894)
Sub-national tax autonomy	0.642*	0.878	0.702*
Ş	(0.169)	(0.103)	(0.137)
Borrowing only for capital investments	1.453	1.222	1.296
	(0.581)	(0.212)	(0.291)
Tax autonomy*Investments	0.571	0.749	0.730
	(0.505)	(0.297)	(0.386)
Sub-national own revenues	0.105	0.390	0.197
	(0.196)	(0.325)	(0.234)
Vertical Fiscal Imbalance	0.026***	0.276***	0.127***
	(0.028)	(0.126)	(0.082)
Administrative*VFI	25.391***	3.086**	6.350***
	(30.216)	(1.585)	(4.520)
Cooperative*VFI	15.017**	2.364	3.735*
	(18.288)	(1.256)	(2.799)
Rule based*VFI	6.741	1.607	2.156
	(9.751)	(1.036)	(1.893)
Market discipline*VFI	2.798	1.231	1.631
	(6.561)	(1.360)	(2.347)
Corruption	1.162	1.083*	1.109
	(0.119)	(0.051)	(0.071)
Government Stability	0.874***	0.951**	0.922***
	(0.041)	(0.020)	(0.028)
GDP per capita growth	0.980	0.989*	0.987
	(0.013)	(0.006)	(0.008)
Population Density	0.986*	0.993**	0.991**
	(0.007)	(0.003)	(0.005)
EU membership	0.452***	0.703***	0.630***

Table 2. Parametric estimation of proportional hazard model

	(0.114)	(0.082)	(0.097)
Duration Dependence Parameter (P)	2.607		
Log likelihood	-689.446	-959.756	-4996.8
AIČ	1424.892	1963.512	10055.6
BIC	1533.951	2067.829	10202.59
Observations	847	847	847

Robust standard errors in parentheses

* significant at 10%; ** significant at 5%; *** significant at 1%

As Table 3 shows, the number of failures under all three *Stronger* definitions is larger than under the Weaker definition. Furthermore, under the *Stronger* definitions, the average probability of ending the fiscal consolidation is higher than under the *Weaker* one, and increases with the threshold, while the average duration decreases with the threshold. The maximum duration under all three *Stronger* definitions is 7 years, with around 68, 72 and 77 percent of consolidations ending after only one year for threshold 0.5, 0.75 and 1 percent, respectively.

Table 3. Descriptive Statistics: Failure and Duration, Alternative definitions of fiscal consolidation

	Threshold 0%	Threshold 0.5%	Threshold 0.75%	Threshold 1%
		Fai	ilure	
Mean	0.390	0.668	0.717	0.760
Standard Deviation	0.488	0.471	0.451	0.428
Variance	0.238	0.222	0.204	0.184
Skewness	0.453	-0.715	-0.962	-1.213
Kurtosis	1.205	1.511	1.925	2.470
Number of failures	330	566	607	643
Observations	847	847	847	847
		Dur	ation	
Mean	2.267	1.542	1.431	1.345
Standard Deviation	1.164	0.972	0.861	0.748
Variance	2.770	0.944	0.740	0.560
Skewness	1.819	2.244	2.619	2.871
Kurtosis	6.742	8.740	11.175	13.428
Observations	847	847	847	847

Figure 4 compares the duration of fiscal consolidations between alternative definitions of fiscal consolidation. Unlike with the *Weak* definition, where 30 percent of fiscal consolidations lasted three or more years, with *Stronger* definitions this percentage decreases to only 13 percent or less.¹¹

¹¹ 13.6 percent for 0.5 threshold; 9.6 percent for 0.75 threshold; 7.3 percent for 1 percent threshold.



Figure 5 and Figure 6 show the Kaplan-Meier survivor and hazard estimates respectively, for all thresholds. Graphs show that the probability of maintaining the consolidation after the first year decreases with larger threshold, while after the second or third year the probability decreases at a slower rate under the *Stronger* definitions than under the *Weak* one. This behavior is translated into a smoothed hazard function that shows higher positive dependence on accumulated duration under the *Stronger* definitions than under the *Weaker* one.



Finally, the original model is re-estimated using the *Stronger* definitions of fiscal consolidation under the new definition of fiscal consolidation. If results obtained under the 1 percent threshold are robust, estimates on explanatory variables will maintain their "signs" and statistical significance. Table 4 presents a comparison of the Weibull estimation under the *Weaker* (threshold 0%) and all three *Stronger* (threshold 0.5%, 0.75% and 1%) definitions of fiscal consolidation.

Table 4 presents results on main variables of interest, sub-national control regimes and selected control variables. Results suggest that with stronger definition of fiscal consolidation, only rule based control and administrative control seem to be consistently more effective in maintaining fiscal discipline than prohibited borrowing at the sub-national level. A surprising result is obtained on market discipline which completely changes sign once a stronger definition of fiscal consolidation is employed. This result may possibly be explained by stronger importance of transparency and governance in imposing tighter requirements in maintaining fiscal consolidation, and in such countries, market discipline is efficient instrument for sub-national borrowing control. Result on government stability is in favor of this preposition. Moreover, with high fiscal dependence on intergovernmental transfers from central government, none of the regimes seem to be efficient in maintaining fiscal consolidation when the applied threshold is higher. This result certainly is in favor of the conjecture that larger sub-national dependence on financing from the central government budget increases the risk of moral hazard, reduces sub-national fiscal responsibility and may result in overspending.



	0.0%	0.5%	0.75%	0.1%	
Number of previous failures	1.146***	0.991	0.977	0.980	
	(0.038)	(0.030)	(0.030)	(0.030)	
Initial budget balance	0.980	0.952**	0.933**	0.917***	
-	(0.028)	(0.023)	(0.025)	(0.029)	
Size of the adjustment	0.007**	0.000	0.000	0.000	
	(0.017)	(0.003)	(0.001)	(0.000)	
Administrative regime	0.306***	0.264**	0.390*	0.381	
	(0.136)	(0.161)	(0.190)	(0.224)	
Cooperative regime	0.343**	0.567	1.095	0.993	
	(0.166)	(0.395)	(0.680)	(0.676)	
Rule based regime	0.230**	0.206**	0.434	0.371*	
	(0.135)	(0.128)	(0.231)	(0.209)	
Market discipline	1.048	0.060**	0.167	0.161	
	(1.536)	(0.079)	(0.225)	(0.272)	
Administrative*VFI	25.391***	14.699	16.648**	15.069	
	(30.216)	(24.426)	(23.669)	(26.255)	
Cooperative*VFI	15.017**	6.104	1.954	1.733	
	(18.288)	(10.943)	(3.138)	(3.069)	
Rule based*VFI	6.741	8.694	17.416*	22.452*	
	(9.751)	(15.859)	(28.882)	(40.090)	
Market discipline*VFI	2.798	56.836*	25.773	106.422	
	(6.561)	(135.319)	(63.607)	(314.500)	
Duration Dependence Parameter (P)	2.607	2.678	2.695	2.709	
Log likelihood	-689.446	-550.965	-493.473	-430.221	
AIC	1424.892	1147.929	1032.945	906.443	
BIC	1533.951	1256.988	1142.004	1015.502	
Observations	847	847	847	847	
Pohust standard errors in parentheses					

Table4. Parametric Weibull Estimation by Threshold

* significant at 10%; ** significant at 5%; *** significant at 1%

VII. Conclusion and Policy Recommendations

This article examined the effect of main four sub-national borrowing control regimes, relative to prohibited borrowing at the sub-national level, on duration of sub-national governments' fiscal consolidations using panel data for 60 countries for the period between 1990 and 2008. Baseline definition for fiscal sustainability in this study assumes threshold of 0 percent and robustness of the results obtained using this definition is tested by increasing the threshold from zero to 1.

The results suggest cooperative regime performs dominantly in sustaining fiscal consolidation, which is somewhat in line with the property of this approach that it combines the advantages off all other three approaches. Second, results suggest that market discipline performs the least favorably comparing to other three approaches and prohibited sub-national borrowing, which corresponds to the fact that in many countries the requirements for market discipline as a successful instrument for sub-national borrowing control are not met. Finally, administrative and cooperative regimes in presence of high fiscal dependence on central government financing seem to increase the probability of ending a consolidation episode. This result is not surprising given that these two regimes refer to significantly high degree of central government control, which may increase the risk of moral hazard. Strong central government control may give encouraging signs to the sub-national governments to over borrow and to expect being bailed out by the central government. This deteriorates the general government budget directly, through unplanned bailout

from the central government, and indirectly, through spillover effect on other sub-national governments who are as well highly fiscally dependent on central government financing. Finally, the robustness analysis suggests that the administrative controls over sub-national borrowing consistently perform dominantly comparing to other regimes. However, this effect vanishes with large central government financing to the sub-national budget.

The results obtained in this study imply following policy recommendations. First, sub-national government borrowing does not have to endanger overall fiscal sustainability if the borrowing control framework is well designed. Second, reducing fiscal dependence on central government financing and, in return, giving more tax autonomy to sub-national government, reduces the risk of moral hazard and improves the effectiveness of borrowing control in maintaining fiscal balance at the sustainable level.

Bibliography

Abrams, B., & Dougan, W. (1986). The effects of institutional restraints on government spending. *Public Choice, 49*, 101–106.

Adam, C., & Bevan, D. (2003). Staying the Course: Maintaining Fiscal Control in Developing Countries. In S. Collins, & D. Rodrik. Brooking Institution Trade Forum 2003.

Ahmad, E., Albino, M. A., & Singh, R. (2005). Subnational Public Financial Management: Institutions and Macroeconomic Considerations. *IMF Working Papers 05/108*.

Ahmad, E., Albino-War, M., & Singh, R. (2005). Subnational Public Financial Management: Institutions and Macroeconomic Considerations. *IMF Working Paper WP/05/108, International Monetary Fund*.

Alibegovic, D. J. (2007). Tackling the Problem of Inadequate Financing for Local Development: The Case of Croatia. In S. Giguère, *Local Innovations for Growth in Central and Eastern Europe*. Paris: OECD.

Alm, J., & Indrawati, S. M. (2004). Decentralization and Local Government Borrowing in Indonesia. In J. Alm, J. Martinez-Vazquez, & S. M. Indrawati, *Reforming intergovernmental fiscal relations and the rebuilding of Indonesia.* Northampton, Massachusetts: Edward Elgar Publishing Limited.

Alt, J., & Lowry, R. C. (1994). Divided government, fiscal institutions and budget deficits: evidence from the states. *American Political Science Review, 88(4)*, 811–828.

Aoki, I. (2008). Decentralization and Intergovernmental Finance in Japan. *PRI Discussion Paper Series No.08A, Research Department Policy Research Institute, MOF Japan*.

Asian Development Bank. (2003). Technical Assistance to the Republic of Indonesia for the Local Government Financing Project.

Bajo, A. (2004). Local Government Unit Borrowing in Croatia: Opportunities and Constraints. *Institute of Public Finance, Croatia*.

Bajo, A. (2007). Assessments of the Effectiveness of the Croatian Fiscal Equalization Model. *Institute of Public Finance and Faculty of Economics and Business, Zagreb*.

Balás, G., & Hegedüs, J. Local Government Borrowing in Hungary. In P. Swianewicz, *Local Government Borrowing: Risks and Rewards*. Budapest: Open Society Institute/LGI.

Belcher, Mark D., Victor Cionga, Francis J. Conway, Petru Filip, Valeriu Ionescu, Gheorghe Ocneanu, Ioan Onisei, George E. Peterson, Juliana H. Pigey, 1997, "Romania: Municipal Creditworthiness and Local Government Decentralization", The Urban Institute, Washington D.C.

Boex, J., Martinez-Vazquez, J., & Schaef, M. (2005). Country Report: An Assessment of Fiscal Decentralization Reform in Georgia. In *Fiscal Decentralization in Transition Economies: Case Studies from the Balkans and Caucasus*. UNDP Bratislava Regional Center.

Boex, J., Martinez-Vazquez, J., & Timofeev, A. (2005). Country Report: An Assessment of Fiscal Decentralization Reform in Armenia. In *Fiscal Decentralization in Transition, Economies: Case Studies from the Balkans and Caucasus.* UNDP Bratislava Regional Center.

Brennan, G., & Buchanan, J. (1980). *The Power to Tax.* New York: Cambridge University Press. Cabasés, F., Pascual, P., & Vallés, J. (n.d.). The effectiveness of institutional borrowing restrictions: empirical evidence from Spanish municipalities. *Public Choice, 131*, 293–313.

Chandler, Mark W.S., 2003, "Revenue Raising and Accountability of Local Governments in Lithuania", paper presented at the 11th NISPAcee Annual Conference, Bucharest, Romania, April 10-12, 2003

Chaparro, Juan Camilo, Michael Smart, Juan Gonzalo Zapata (2005), Intergovernmental Transfers and, Municipal Finance in Colombia, in ed. Richard M. Bird, Joel Slemrod, James M. Poterba, Fiscal Reform in Colombia: Problems and Prospects, MIT Press, Cambridge, Massachusetts

Council of Europe . (2008). Structure and Operation of Local and Regional Democracy in Denmark. Council of Europe. (1996). Structure and Operation of Local and Regional Democracy in Lithuania.

Council of Europe. (1996). Structure and Operation of Local and Regional Democracy in Sweden.

Council of Europe. (1997). Structure and Operation of Local and Regional Democracy in Denmark.

Council of Europe. (1997). Structure and Operation of Local and Regional Democracy in France.

Council of Europe. (1997). Structure and Operation of Local and Regional Democracy in Iceland.

Council of Europe. (1997). Structure and Operation of Local and Regional Democracy in Norway.

Council of Europe. (1998). Structure and Operation of Local and Regional Democracy in Germany.

Council of Europe. (1998). Structure and Operation of Local and Regional Democracy in Ireland.

Council of Europe. (1999). Structure and Operation of Local and Regional Democracy in Austria.

Council of Europe. (1999). Structure and Operation of Local and Regional Democracy in United Kingdom.

Council of Europe. (2000). Structure and Operation of Local and Regional Democracy in Greece.

Council of Europe. (2006). Structure and Operation of Local and Regional Democracy in Iceland.

Council of Europe. (2006). Structure and Operation of Local and Regional Democracy in Latvia.

Council of Europe. (2006). Structure and Operation of Local and Regional Democracy in Lithuania.

Council of Europe. (2006). Structure and Operation of Local and Regional Democracy in Portugal.

Council of Europe. (2009). Structure and Operation of Local and Regional Democracy in Finland.

Dabla-Norris, E. (2005). Issues in Intergovernmental Fiscal Relations in China. International Monetary Fund .

Dabla-Norris, E. (2006). The Challenge of Fiscal Decentralisation in Transition Countries. *Comparative Economic Studies*, 48, 100–131.

Dabla-Norris, E., & Wade, P. (2002). The Challenge of Fiscal Decentralization in Transition Countries. IMF

Working Paper WP/02/103.

Danielian, A. (2002). Local Government in Russia: Reinforcing Fiscal Autonomy. In K. Davey, *Fiscal Autonomy and Efficiency: Reforms in the Former Soviet Union, Local Government and Public Service Reform Initiative.* Budapest, Hungary: Open Society Institute.

De Mello, L. (2001). Fiscal decentralization and borrowing costs: the case of local governments. *Public Finance Review 29(2)*, 108-138.

De Mello, L. R. (2000). Fiscal Decentralization and Intergovernmental Fiscal Relations: A Cross-Country Analysis. *World Development, Vol. 28, No. 2*, 365-380.

De Mello, Luiz, 2007, Fiscal Responsibility Legislation and Fiscal Adjustment: The Case of Brazilian Local Governments, in ed. George E. Peterson and Patricia Clarke Annez, Financing Cities: Fiscal Responsibility and Urban Infrastructure in Brazil, China, India, Poland and South Africa, The World Bank

Derycke, P., & Gilbert, G. (1985). The public debt of French local government. *Journal of Public Policy, 5(3)*, 387–399.

Dillinger, William, 2007, Intergovernmental Fiscal Relations in the New EU Member States, Consolidating Reforms, World Bank Working Paper No. 111

Dilliniger, William, and Steven B. Webb, 1999, Decentralization and Fiscal Management in Colombia, Policy Research Working Paper 2122, The World Bank

Dunn, Jonathan & Deborah Wetzel (1999) Fiscal Decentralization in The ECA Region: Progress and Prospects, The World Bank

Eaton, Kent and J. Tyler Dickovick (2004), The Politics of Re-Centralization in Argentina and Brazil, Latin American Research Review, Vol. 39, No. 1 (2004), pp. 90-122

Feld, L. P., Kirchgässner, G., & Schaltegger, C. A. (2003). Decentralized Taxation and the Size of Government: Evidence from Swiss State and Local Governments. *CESifo Working Paper No. 1087*. Fenge, R., & Meier, V. (2002). Why cities should not be subsidized. *Journal of Urban Economics*, 433-447.

Fornasari, F., & Steven, W. (2000). The Macroeconomic Impact of Decentralized Spending and Deficits: International Evidence. *Annals of Economics and Finance, Vol. 1*, 404-433.

Freinkman, Lev and Plamen Yossifov, (1999), Decentralization in Regional Fiscal Systems in Russia, Trends and Links to Economic Performance, Policy Research Working Paper WPS 2100, The World Bank

Fukasaku, K., & De Mello, L. (1998). Fiscal Decentralization and Macroeconomic Stability: The Experience of Large Developing and Transition Economies. In. In K. Fukasaku, & R. Hausmann, *Democracy, Decentralization and Deficits in Latin America*. Paris: OECD.

Garcia-Milà, Teresa, and Therese J. McGuire, 2002, Fiscal Decentralization in Spain: An Asymmetric Transition to Democracy, Department of Economics and Business, Universitat Pompeu Fabra in its series Economics Working Papers No. 866

Garg, Subhash Chandra , 2007, Overview of Urban Infrastructure Finance in India, in ed. George E. Peterson, Patricia Clarke Annez, Financing Cities: Fiscal Responsibility and Urban Infrastructure in Brazil, China, India, Poland and South Africa, The Worl Bank

George, K., & Symansky, S. A. (1998). Fiscal Policy Rules. International Monetary Fund .

German Advisory Group, (2004). Local Government Revenues: Increasingly Dependent on Kyiv?, Insitute for Economic Research and Polocy Consulting in Ukraine

Giugale, Marcelo, Fausto Hernandez Trillo, and João C. Oliveira, 2000, Subnational Borrowing and Debt Management in "Achievements and Challenges of Fiscal Decentralization Lessons from Mexico" Edited by Marcelo M. Giugale and Steven B. Webb, World Bank 2000

Giuriato, L., & Gastaldi, F. (2009). The domestic stability pact in Italy: a rule for discipline? *MPRA Paper with number 15183*.

Glasser, Matthew D., and Charles Jokay, 2000, Advisory Note on Regulatory Framework and Institutional Issues: Municipal Borrowing in Bosnia and Herzegovina, Local Development Pilot Project Bosnia and Herzegovina

Hall, Luis J. , Gilberto E. Arce, and Alexander Monge N., 2002, Bailouts in Costa Rica as a result of Government Centralization and Discretionary Transfers, Research Network Working paper #R-475, Inter-American Development Bank

Haug, P. (2004). Decreasing Population and Rising Costs of Providing Water and Sewage Treatment within Cities: A Case Study. *ERSA Conference Papers. Vienna, Austria: European Regional Science Association*.

Hawkesworth, I., Bergvall, D., Emery, R., & Wehner, J. (2008). Budgeting in Greece. OECD Journal on Budgeting .

Hayek, F. (1945). The use of knowledge in society. American Economic Review 35, 519-530.

Heredia-Ortíz, E., & Rider, M. (n.d.). India's Intergovernmental Transfer System and the Fiscal Condition of the States. *Andrew Young School of Policy Studies, Research Paper Series, Department of Economics, International Studies Program, Working Paper 06-47*.

Hunter, J., & Shah, A. (1996). A simple measure of good governance and its application to the debate on the appropriate level of fiscal decentralization. *Worl Bank*.

IMF, 2006, Panama: Selected Issues and Statistical Appendix, 2006, IMF Country Report No. 06/3, International Monetary Fund

Internatinal Monetary Fund. (2001). The Government Finance Statistics Manual 2001.

International Monetary Fund. (2001). Belgium, Selected Issues. IMF Country Report No. 01/45.

International Monetary Fund. (2006). Bolivia: Selected Issues. IMF Country Report No. 06/273.

International Monetary Fund. (2008). Austria: Selected Issues. IMF Country Report No. 08/189.

Jaanso, A., Liivik, S., Jogi, A., & Milt, T. (2004). Local Government Borrowing: Regulations and Practices in Estonia. In P. Swianewicz, *Local Government Borrowing: Risks and Rewards.* Budapest: Open Society Institute/LG.

Ježek, Vladimír, Hana Marková and Lukáš Vána, 2004, Local Government Borrowing in the Deregulated Market of the Czech Republic, in ed. Pawel Swianewicz, Local Government Borrowing: Risks and Rewards, Budapest: Open

Society Institute/LGI.

Kenyon, D. (1991). Effects of federal volume caps on state and local borrowing. *National Tax Journal, 44*, 81–92.

Kim, Junghun. (2003), Republic of Korea, in ed. Yun-Hwan Kim, Local Government Finance and Bond Markets, Asian Development Bank

Kling, J., & Viktor Nižnanský. (2004) Local Borrowing in Slovakia: From Deregulation to Regulation and Stabilization, in ed. Pawel Swianewicz, Local Government Borrowing: Risks and Rewards, Budapest: Open Society Institute/LGI.

Klun, M. (2006) High level of centralization and local debt in Slovenia. http://www.nispa.sk/_portal/files/conferences/2006/papers/200606011217410.Klunfinalpaper.pdf

Kopanska, A., & Levitas, T. (2004). The Regulation and Development of the Subsovereign Debt Market in Poland: 1993–2002. In P. Swianewicz, *Local Government Borrowing: Risks and Rewards.* Budapest: Open Society Institute/LGI.

Koutsogeorgopoulou, Vassiliki , 2007, Fiscal Relations Across Levels of Government in Australia, Economics Department Working Papers No. 541, OECD

Kovacs, Gabor. (2007) Financing local investments by issuing municipal bonds. <u>http://www.nispa.sk/_portal/files/conferences/2007/papers/200704110019070. MunicipalBondsinHungary.doc</u>

Lane, T. (1993). Market Discipline. IMF Staff Papers 40, 53-88.

Laubach, Thomas (2005), "Fiscal Relations Across Levels of Government in the United States", OECD Economics Department Working Papers, No. 462

Leprince, Matthieu, Thierry Madies, and Sonia Paty, 2007, Business Tax Interactions among Local Governments: An Empirical Analysis of the French Case, JOURNAL OF REGIONAL SCIENCE, VOL. 47, NO. 3, 2007, pp. 603– 621

Leschenko, Natalie, and Manuela Troschke, 2006, Fiscal Decentralization in Centralized States: The Case of Central Asia, Osteuropa-Institut Münichen, Working Papers No. 261

Liebig, Klaus , Mandana Bahrinipour, Laura Fuesers, Benjamin Knödler, Christian Schönhofen, Mareike Stein, (2008), Municipal Borrowing for Infrastructure Service Delivery in South Africa – a Critical Review, Studies Deutsches Institut für Entwicklungspolitik, Bonn

Lin, D. Y., & Wei, L. J. (1989). The robust Inference for the Cox Proportional Hazard Model. *Journal of American Statistical Association 84*, 37-47.

Martell, C. (2008). Fiscal Institutions of Brazilian Municipal Borrowing. *Public Administration and Development 28*, 30-41.

Martell, C. R., & Guess, G. M. (2006). Development of local government debt financing markets: Application of a market-based framework. *Public Budgeting and Finance*, 88–119.

Martinez-Vazquez, J., & Boex, J. (2001). Russia's Transition to a New Federalism. World Bank Institute.

Mikayilov, E. (2007). Intergovernmental Fiscal Transfers in Azerbaijan: Role of Tax - Sharing in Local Government Financing. *Economic Research Center, Central European Initiative*.

Ming, Su, and Zhao Quanhou, 2007, China: Fiscal Framework and Urban Infrastructure Finance, in ed. George E. Peterson and Patricia Clarke Annez, Financing Cities: Fiscal Responsibility and Urban Infrastructure in Brazil, China, India, Poland and South Africa, The World Bank

Ministry of Finance. 2005. Local Government in Sweden. Organisation, Activities and Finance. Stockholm: Regeringskansliet

Mochida, Nobuki , 2001, Taxes and Transfers in Japan's Local Public Finances, World Bank Institute

Mochida, Nobuki, 2008, Subnational Borrowing in Japan: from 'Implicit Guarantee' to Market

Discipline and Fiscal Rule, Subnational Borrowing in Japan: from 'Implicit Guarantee' to Market Discipline and Fiscal Rule, CIRJE Discussion Papers F-541 (http://www.e.u-tokyo.ac.jp/cirje/research/03research/02dp.html)

Moges, A. G. (2003). An Economic Analysis of Fiscal Federalism in Ethiopia. *Northeast African Studies, Volume 10, Number 2*, 111-140.

Mora, Mônica, and Ricardo Varsano, 2001, Fiscal Decentralization and Sub-national Fiscal Autonomy in Brazil: Some Facts of the Nineties, Discussion Paper No. 854, The Instituto de Pesquisa Econômica Aplicada (IPEA), Brazil

Musgrave, R. A. (1959). The Theory of Public Finance. New York: McGraw Hill.

Nikiforov, Sergey Mikhaylovich, Andrey Vasil'evich Cherniavsky, Vladislav Eduardovich Grigorov, Igor Vyacheslavovich Belyakov and Konstantin Gur'evich Tioussov, 2004, Local Government Borrowing in Russia: Difficult Paths of Economic Transformation, in ed. Pawel Swianewicz, Local Government Borrowing: Risks and Rewards, Budapest: Open Society Institute/LGI.

Nikolov, Marjan, 2006, The Future of Local Government Finance: Case studies from Romania, Bulgaria and Macedonia, Center for Economic Analysis, Skopje

Oates, W. (1972). Fiscal Federalism. New York: Harcourt Brace Jovanovich.

Oates, W. (1993). Fiscal Decentalization and Economic Development. . *National Tax Journal 46*, 237-243. OECD 2003 Fiscal Relations across Levels of Government

OECD 2003, Territorial Reviews: Helsinki Finland, OECD 2003

OECD, 2003, Economic Outlook: December 2003 No. 74 - Fiscal relations across levels of government

OECD, 2003, Economic Surveys: Ireland - Volume 2003 Issue 9

OECD, 2006, Economic Surveys: Iceland - Volume 2006 Issue 14

OECD, 2007, Economic Surveys: Belgium, Volume 2007/3, 2007

OECD, 2008, Territorial Reviews Portugal 2008

OECD, 2009, Territiorial Review: Chile, OECD 2009

OECD. (2006). Intergovernmental Transfers for Environmental Infrastructure, Lessons from Armenia, the Russian Federation and Ukraine.

O'Neill, Kathleen (2006), Who's Tugging at the Purse Strings? The Political Economy of Intergovernmental Transfers in Argentina, Mexico and Peru, in ed. Paul Smoke, Eduardo J. Gomez, George E. Peterson, Decentralization in Asia and Latin America, Towards a Comparative Interdisciplinary Perspective, Edward Elgar, Cheltenham, UK; Northampton MA, USA

Peterson, G. (2000). Building local credit systems. *Urban and Local Government Background Series 3, World Bank, Washington, DC*.

Pigey, Juliana (1999) Fiscal Decentralization and Local Government Finance in Hungary 1989–1999, Urban Institute, USAID

Plekhanov, A., & Singh, R. (2006). How should subnational government borrowing be regulated? Some crosscountry evidence. *IMF Staff Papers Vol. 53, No. 3*, 426–452.

Poterba, J. M. (1995). Balanced budget rules and fiscal policy: evidence from the states. *National Tax Journal, 48*, 329–336.

Prud'homme, R. (1994). The Dangers of Decentralization. World Bank Research Observer, Volume 10, Number 3.

Purfield, C. (2004). The Decentralization Dilemma in India. IMF Working Paper WP/04/32.

Rao, M. Govinda, and Nirvikar Singh, 2001, The Political Economy of Center-State Fiscal Transfers in India, Working Paper No. 107, Center for Research on Economic Development and Policy Reform, Stanford University

Rattso, Jorn. (2005) Local tax financing in the Nordic countries, Economic General Report, in ed. Robert Pahlsson, Yearbook for Nordic Tax Research

Reid, B. G. (2203). Fiscal Management and Stabilization Policy in Federal Systems: The Experience of Argentina and Mexico. In P. Boothe, *Fiscal Relations in Four Countries: Four Essays.* Ottawa: Forum of Federations.

Republic of Serbia, 2006, Zakon o finansiranju lokalne samouprave 2006.

Republic of Serbia. (2005). Zakon o javnom dugu 2005.

Rodden, J. (2002). The Dilemma of Fiscal Federalism: Grants and Fiscal Performance around the World. *American Journal of Political Science*, 670-687.

SALAR, 2006, The Creditworthiness of Swedish local governments, Stockholm 2006, Swedish Association of Local Authorities and Regions; www.kommuninvest.se/trycksaker/ovrigt. Shah, A. (2005). Fiscal Decentralization and Fiscal Performance. *World Bank Policy Research Working Paper 3786*.

Shome, P. (2002). India's Fiscal Matters. New Delhi: Oxford University Press.

Singh, U. B. 2009 Decentralized democratic governance in new millennium : local government in the USA, UK, France, Japan, Russia and India, Concept Pub. Co., New Delhi

Stein, E. (1999). Fiscal Decentralization and Government Size in Latin America. *Journal of Applied Economics, Vol. 2, No. 2.*

Tanzi, V. (2000). On Fiscal Federalism; Issues to Worry About. *Conference on Fiscal Decentralization, International Monetary Fund*.

Ter-Minassian, T. (1997). *Fiscal Federalism in Theory and Practice.* Washington D.C.: International Monetary Fund.

Ter-Minassian, T., & Craig, J. (1997). Controls of Subnational Government Borrowing. In T. Ter-Minassian, *Fiscal Federalism in Theory and Practice* (pp. 156-172). Washington D.C.: International Monetary Fund.

Ter-Minassian, T., & Craig, J. (1997). Controls of Subnational Government Borrowing. In T. Ter-Minassian, *Fiscal Federalism in Theory and Practice* (pp. 156-175). Washington DC: IMF.

The Urban Institute, 2007, "Local Government and Decentralization in Albania - Completion Report July 2004 to July 2007", UI Project 06901-020, August 2007

Thoni, E., Garbislander, G., & Haas, D. (2002). Local Budgeting And Local Borrowing In Austria. In B. Dafflon, *Local Public Finance in Europe* (pp. 45-74). Cheltenham (UK); Northampton (USA): E.Elgar.

Tiebout, C. (1956). A Pure Theory of Local Expenditure. Journal of Political Economy 64, 416-424.

Toboso, Fernando (2007), Some key organisational rules for financing regional governments in democratic Spain and their impacts. EGPPA Annual Conference, September 19th-22th, 2007, Madrid.

Toonen, Theo, Trui Steen, Frits van de Meer, 2007, Local Government Finance and Intergovernmental Relations: Comparative Institutional Analysis, Leiden University, Department of Public Administration

Trasberg, Viktor. (2004) Intergovernmental fiscal relations in the Baltic municipalities. <u>http://unpan1.un.org/intradoc/groups/public/documents/nispacee/unpan009153.pdf</u>

Webb, S. (2004). Fiscal Responsibility Laws for Subnational Discipline: The Latin American Experience. *World Bank Policy Research Working Paper No. 3309*.

Wescott, C. G. (2006). Decentralization Policy and Practice in Vietnam: 1991 - 2003. In P. Smoke, E. J. Gomez, & G. E. Peterson, *Decentralization in Asia and Latin America, Towards a Comparative Interdisciplinary Perspective.* Cheltenham, UK; Northampton, MA, USA: Edward Elgar.

Wibbels, E., & Rodden, J. (2007). Fiscal Decentralization and the Business Cycle: An Empirical Study of Eight Federations. *Proceedings of the Conference New Perspectives on Fiscal Federalism: Intergovernmental Relations, Competition and Accountability.* Berlin: Social Science Research Center.

World Bank, 1997, Belarus: prices, markets, and enterprise reform, Country Study 16863

World Bank, 2007Spending for Development: Making the Most of Indonesia's New Opportunities, Indonesia Public Expenditure Review 2007, The World Bank Office Jakarta

World Bank, 2008, Poverty Reduction and Economic Management Unit, Europe and Central Asia Region, "Albania Local Finance Policy Note Programmatic Public Expenditure and Institutional Review", World Bank, Report No.

44109-AL, September 2, 2008

Zigiene, G., Ciburiene, J., & Chandler, M. W. (2006). Local Government Finance in Lithuania. In Z. Ševic, *Local Public Finance In Central And Eastern Europe Head, Division of Accounting, Finance and Risk.* Caledonian Business School, Glasgow Caledonian University, UK.

Appendix

Table A1. Descriptive Statistics

Variable	Obs	Mean	Std. Dev.	Min	Max
Prohibited	847	0.17	0.38	0.00	1.00
Administrative	847	0.39	0.49	0.00	1.00
Cooperative	847	0.14	0.35	0.00	1.00
Rule Based	847	0.12	0.33	0.00	1.00
Market Discipline	847	0.17	0.38	0.00	1.00
Sub-national Tax Autonomy	847	0.57	0.49	0.00	1.00
Borrowing for only investment purpose	847	0.44	0.50	0.00	1.00
General Government Primary Balance	847	0.04	0.08	-0.32	0.75
Central Government Primary Balance	847	0.02	0.06	-0.36	0.37
Sub-national Own Revenues	847	0.15	0.12	0.00	0.69
Vertical Fiscal Imbalance	847	0.39	0.21	0.01	0.89
GDP per capita growth rate	847	3.11	4.01	-21.17	14.02
Population density	847	106.57	105.56	2.22	492.32
Corruption	847	3.61	1.44	0.33	6.00
Government Stability	847	8.31	1.71	2.92	12.00

Table A2. Sample structure by sub-national borrowing control regime

	Number of observations	Percentage	Number of id*
Prohibited	146	17.24	16
Administrative	331	39.08	30
Cooperative	119	14.05	7
Rule Based	103	12.16	10
Market Discipline	148	17.47	9
Total	847	100	

* Does not add to 60 because countries change dominant control regime over the observed period

Table A3. Variable description and sources

Variable Description		Source	
Prohibited	=1 if sub-national government prohibited		
Administrative	=1 if administrative is dominant sub- national borrowing control regime		
Cooperative	=1 if cooperative is dominant sub- national borrowing control regime		
Rule Based	=1 if rule based is dominant sub-national borrowing control regime	Various sources	
Market Discipline	=1 if market discipline is dominant as sub-national borrowing control regime		
Sub-national Tax Autonomy	=1 if sub-national government has authority to set tax rates and/or choose tax base		
Borrowing for only investment	=1 if sub-national borrowing is allowed only for investment purposes		
General Government Primary Balance	General Government: Revenue - Expenditures + Interest Payments		
Central Governmetn Primary Balance	Central Government: Revenue - Expenditures + Interest Payments	Government Finance Statistics Database, International Monetary Fund; Various individual country sources (See Table A4 for details)	
Sub-national Own Revenues Vertical Fiscal Imbalance	Share of sub-national own revenues in total general government revenues Share of sub-national intergovernmental transfers received from the central government in total sub-national revenues		
GDP per capita growth rate	Annual nominal GDP per capita growth rate	World Development Indicators	
Population density	Population per square kilometer		
Corruption	Assessment of corruption within the political system	International Country Disk Guide	
Government Stability	Assessment both of the government's ability to carry out its declared program(s), and its ability to stay in office	The PRS Group	

id	Country	Observed Period	Government Finance	Qualitative Indicators
			Fiscal Statistics of Government, 2008, Ministry of Finance,	Dabla-Norris (2006); Urban
1	Albania	2000 - 2008	Albania	Institute (2007); World Bank (2008)
2	Argentina	1990 - 2004	IMF GFS	Reid (2003); O'Neill (2006)
3	Armenia	2002 - 2008	IMF GFS	Boex et al. (2005); OECD (2006)
4	Australia	1990 - 2008	IMF GFS	Koutsogeorgopoulou (2007)
				Council of Europe (1999); OECD
5	Austria	1990 - 2008	IMF GFS	(2003), Fuerites et al. (2000), INF (2008)
6	Azerbaijan	1994 - 1999	IMF GFS	Mikayilov (2007)
7	Belarus	1992 - 2008	IMF GFS	World Bank (1997)
8	Belgium	1990 - 2007	IMF GFS	IMF (2001); OECD (2007)
9	Bolivia	1990 - 2008	IMF GFS	IMF (2006)
10	Bosnia and Herzegovina	2004 - 2008	IMF GFS	Glasser et al. (2000)
				Mora and Varsano (2001); Eaton
11	Brazil	1990 - 2008	IME GES: Ministry of Finance Brazil (http://www.fazenda.gov.br/)	(2007)
12	Bulgaria	1990 - 2008	IME GES	Nikolov (2006)
13	Canada	1990 - 2008	IMF GFS	OECD (2003)
14	Chile	1991 - 2008	IMF GFS	OECD (2009)
15	China	1995 - 2008	IMF GFS	Dabla-Norris (2005)
				Dillinger and Webb (1999):
16	Colombia	1998 - 2008	IMF GFS	Chaparro et al. ((2005)
17	Costa Rica	2001 - 2007	IMF GFS	Hall et al. (2002)
				Bajo (2004); Bajo (2007);
18	Croatia	1994 - 2008	IMF GFS	Alibegovic (2007)
19	Czech Republic	1993 - 2008	IMF GFS	Jezek et al. (2004); Dillinger (2007)
				Council of Europe (1997, 2008);
20	Denmark	1990 - 2008	IMF GFS	Rattso (2005)
21	El Salvador	2001 - 2008	IMF GFS	IADB (1997)
22	Estonia	1995 - 2008	IMF GFS	Janso et al. (2004); Trasberg (2004); Dillinger (2007)
23	Finland	1990 - 2008	IMF GFS	Council of Europe (1997, 2009); OECD (2003)

Table A4. Suurces on	Government Finance a	nd Oualitative Indicators	by Country

				Council of Europe (1997), Leprince
24	France	1990 - 2008	IMF GFS	(2007)
25	Georgia	1996 - 2007	IMF GFS	Boex et al. (2005)
26	Germany	1990 - 2008	IMF GFS	Council of Europe (1998)
				Council of Europe (2000);
27	Greece	1994 - 2007	IMF GFS	Hawkesworth (2008)
28	Honduras	2002 - 2008	IMF GFS	IADB (1997)
29	Hungary	1990 - 2008	IMF GFS	Pigey (1999), Balas et al. (2004); Kovacs (2007)
30	Iceland	1990 - 2008	IMF GFS	Council of Europe (1997, 2006); OECD (2006)
31	India	1990 - 2008	IMF GFS; Ministry of Finance India (http://finmin.nic.in/)	Rao and Singh (2001), Purfield (2004); Heredia-Ortiz and Rider (2005); Garg (2007)
32	Indonesia	1990 - 2008	IMF GFS; Central Bureau of Statistics of the Republic of Indonesia (http://www.bps.go.id/aboutus.php?tabel=1&id_subyek=13)	ADB (2003); Alm and Indrawati (2004); World Bank (2007)
33	Ireland	1990 - 2007	IMF GFS	Council of Europe (1998); OECD (2003)
34	Italy	1994 - 2008	IMF GFS	Giuriato and Gastaldi (2009)
35	Japan	2001 - 2007	IMF GFS; Japan Statistical Yearbook, various years	Mochida (2001), Aoki (2008), Mochida (2008)
36	Kazakhstan	1997 - 2008	IMF GFS	Leschenko and Troschke (2006)
37	Korea	2005 - 2008	IMF GFS; Ministry of Strategy and Finance, South Korea	Kim (2003), OECD (2003)
38	Latvia	1994 - 2008	IMF GFS	Dunn and Wetzel (1999); Viktor (2004); Council of Europe (2006)
39	Lithuania	1991 - 2008	IMF GFS	Council of Europe (1996, 2006), Zigiene et al. (2008)
40	Macedonia	2005 - 2008	IMF GFS	Nikolov (2006)
41	Mexico	1990 - 2000	IMF GFS	Giugale et al. (2000)
42	Netherlands	1990 - 2008	IMF GFS	Kopits and Symansky (1998)
				Council of Europe (1007) Pattso

			IMF GFS; Pakistan Statistical Yearbook, various years, Federal	
44	Pakistan	1997 - 2008	Bureau of Statistics	Bahl, Wallace and Cyan (2008)
45	Panama	1990 - 1994	IMF GFS	IMF (2006)
46	Poland	1993 - 2008	IMF GFS	Kopanska and Levitas (2004); Dillinger (2007)
47	Portugal	1990 - 2007	IMF GFS	Council of Europe (2006); OECD (2008)
48	Romania	1990 - 2008	IMF GFS	Belcher et al. (1997); Nikolov (2006)
49	Russian Federation	1998 - 2008	IMF GFS	Freinkman and Yossifov (1999); Danielian (2002); Nikiforov et al. (2004); Singh (2009)
			Bulletin, Public Finance, December 2009, Ministry of Finance	
50	Serbia	2002 - 2008	Serbia (http://www.mfin.sr.gov.yu/eng/)	Republic of Serbia (2005, 2006)
51	Slovakia	1996 - 2008	IMF GFS	Kling and Niznansky (2004)
52	Slovenia	1992 - 2007	IMF GFS	Klun (2006)
53	South Africa	1990 - 2007	IMF GFS	Liebig et al. (2008)
54	Spain	1990 - 2008	IMF GFS	Garcia-Mila and McGuire (2002); Ahmad et al. (2005); Toboso (2007)
55	Sweden	1990 - 2001	IME GES	Council of Europe (1996); Rattso (2005); Ministry of Finance Sweden (2005); SALAR (2006), Toopen et al. (2007)
56	Switzerland	1990 - 2007	IME GES	Feld et al. (2003)
				German Advisory Group (2004),
57	Ukraine	1998 - 2008	IMF GFS	OECD (2006)
58	United Kingdom	1990 - 2008	IMF GFS	Council of Europe (1999), OECD (2003)
59	United States	1990 - 2001	IMF GFS	Thomas (2005)
			Ministry of Finance, Vietnam	
60	Vietnam	2002 - 2008	(http://www.mof.gov.vn/DefaultE.aspx?tabid=5740)	Wescott (2006)

2009

2009/1. Rork, J.C.; Wagner, G.A.: "Reciprocity and competition: is there a connection?"

2009/2. Mork, E.; Sjögren, A.; Svaleryd, H.: "Cheaper child care, more children"

2009/3. Rodden, J.: "Federalism and inter-regional redistribution"

2009/4. Ruggeri, G.C.: "Regional fiscal flows: measurement tools"

2009/5. Wrede, M .: "Agglomeration, tax competition, and fiscal equalization"

2009/6. Jametti, M.; von Ungern-Sternberg, T.: "Risk selection in natural disaster insurance" 2009/7. Solé-Ollé, A; Sorribas-Navarro, P.: "The dynamic adjustment of local government budgets: does Spain behave differently?"

2009/8. Sanromá, E.; Ramos, R.; Simón, H.: "Immigration wages in the Spanish Labour Market: Does the origin of human capital matter?"

2009/9. Mohnen, P.; Lokshin, B.: "What does it take for and R&D incentive policy to be effective?"

2009/10. Solé-Ollé, A.; Salinas, P..: "Evaluating the effects of decentralization on educational outcomes in Spain?"

2009/11. Libman, A.; Feld, L.P.: "Strategic Tax Collection and Fiscal Decentralization: The case of Russia"

2009/12. Falck, O.; Fritsch, M.; Heblich, S.: "Bohemians, human capital, and regional economic growth"

2009/13. Barrio-Castro, T.; García-Quevedo, J.: "The determinants of university patenting: do incentives matter?" 2009/14. Schmidheiny, K.; Brülhart, M.: "On the equivalence of location choice models: conditional logit, nested logit and poisson"

2009/15. Itaya, J., Okamuraz, M., Yamaguchix, C.: "Partial tax coordination in a repeated game setting"

2009/16. Ens, P.: "Tax competition and equalization: the impact of voluntary cooperation on the efficiency goal"

2009/17. Geys, B., Revelli, F.: "Decentralization, competition and the local tax mix: evidence from Flanders"

2009/18. Konrad, K., Kovenock, D.: "Competition for fdi with vintage investment and agglomeration advantages" 2009/19. Loretz, S., Moorey, P.: "Corporate tax competition between firms"

2009/20. Akai, N., Sato, M.: "Soft budgets and local borrowing regulation in a dynamic decentralized leadership model with saving and free mobility"

2009/21. Buzzacchi, L., Turati, G.: "Collective risks in local administrations: can a private insurer be better than a public mutual fund?'

2009/22. Jarkko, H.: "Voluntary pension savings: the effects of the finnish tax reform on savers' behaviour"

2009/23. Fehr, H.; Kindermann, F.: "Pension funding and individual accounts in economies with life-cyclers and myopes"

2009/24. Esteller-Moré, A.; Rizzo, L.: "(Uncontrolled) Aggregate shocks or vertical tax interdependence? Evidence from gasoline and cigarettes"

2009/25. Goodspeed, T.; Haughwout, A.: "On the optimal design of disaster insurance in a federation"

2009/26. Porto, E.; Revelli, F.: "Central command, local hazard and the race to the top"

2009/27. Piolatto, A.: "Plurality versus proportional electoral rule: study of voters' representativeness"

2009/28. Roeder, K .: "Optimal taxes and pensions in a society with myopic agents"

2009/29, Porcelli, F.: "Effects of fiscal decentralisation and electoral accountability on government efficiency evidence from the Italian health care sector"

2009/30, Troumpounis, O.: "Suggesting an alternative electoral proportional system. Blank votes count"

2009/31, Mejer, M., Pottelsberghe de la Potterie, B.: "Economic incongruities in the European patent system"

2009/32, Solé-Ollé, A.: "Inter-regional redistribution through infrastructure investment: tactical or programmatic?"

2009/33, Joanis, M.: "Sharing the blame? Local electoral accountability and centralized school finance in California" 2009/34, Parcero, O.J.: "Optimal country's policy towards multinationals when local regions can choose between firm-specific and non-firm-specific policies"

2009/35, Cordero, J.M.; Pedraja, F.; Salinas, J.: "Efficiency measurement in the Spanish cadastral units through DEA"

2009/36, Fiva, J.; Natvik, G.J.: "Do re-election probabilities influence public investment?"

2009/37, Haupt, A.; Krieger, T.: "The role of mobility in tax and subsidy competition"

2009/38, Viladecans-Marsal, E; Arauzo-Carod, J.M.: "Can a knowledge-based cluster be created? The case of the Barcelona 22@district"

2010

2010/1, De Borger, B., Pauwels, W.: "A Nash bargaining solution to models of tax and investment competition: tolls and investment in serial transport corridors"

2010/3, Esteller-Moré, A.; Rizzo, L.: "Politics or mobility? Evidence from us excise taxation"

2010/4, Roehrs, S.; Stadelmann, D.: "Mobility and local income redistribution"

^{2010/2,} Chirinko, R.; Wilson, D.: "Can Lower Tax Rates Be Bought? Business Rent-Seeking And Tax Competition Among U.S. States"

2010/5, Fernández Llera, R.; García Valiñas, M.A.: "Efficiency and elusion: both sides of public enterprises in Spain"

2010/6, González Alegre, J.: "Fiscal decentralization and intergovernmental grants: the European regional policy and Spanish autonomous regions"

2010/7, Jametti, M.; Joanis, M.: "Determinants of fiscal decentralization: political economy aspects"

2010/8, Esteller-Moré, A.; Galmarini, U.; Rizzo, L.: "Should tax bases overlap in a federation with lobbying?"

2010/9, Cubel, M .: "Fiscal equalization and political conflict"

2010/10, Di Paolo, A.; Raymond, J.L.; Calero, J.: "Exploring educational mobility in Europe"

2010/11, Aidt, T.S.; Dutta, J.: "Fiscal federalism and electoral accountability"

2010/12, Arqué Castells, P.: "Venture capital and innovation at the firm level"

2010/13, García-Quevedo, J.; Mas-Verdú, F.; Polo-Otero, J.: "Which firms want PhDS? The effect of the university-industry relationship on the PhD labour market"

2010/14, Calabrese, S.; Epple, D.: "On the political economy of tax limits"

2010/15, Jofre-Monseny, J.: "Is agglomeration taxable?"

2010/16, Dragu, T.; Rodden, J.: "Representation and regional redistribution in federations"

2010/17, Borck, R; Wimbersky, M.: "Political economics of higher education finance"

2010/18, Dohse, D; Walter, S.G.: "The role of entrepreneurship education and regional context in forming entrepreneurial intentions"

2010/19, Åslund, O.; Edin, P-A.; Fredriksson, P.; Grönqvist, H.: "Peers, neighborhoods and immigrant student achievement - Evidence from a placement policy"

2010/20, Pelegrín, A.; Bolance, C.: "International industry migration and firm characteristics: some evidence from the analysis of firm data"

2010/21, Koh, H.; Riedel, N.: "Do governments tax agglomeration rents?"

2010/22, Curto-Grau, M.; Herranz-Loncán, A.; Solé-Ollé, A.: "The political economy of infraestructure construction: The Spanish "Parliamentary Roads" (1880-1914)"

2010/23, Bosch, N.; Espasa, M.; Mora, T.: "Citizens' control and the efficiency of local public services"

2010/24, Ahamdanech-Zarco, I.; García-Pérez, C.; Simón, H.: "Wage inequality in Spain: A regional perspective" **2010/25, Folke, O.:** "Shades of brown and green: Party effects in proportional election systems"

2010/26, Falck, O.; Heblich, H.; Lameli, A.; Südekum, J.: "Dialects, cultural identity and economic exchange"

2010/27, Baum-Snow, N.; Pavan, R.: "Understanding the city size wage gap"

2010/28, Molloy, R.; Shan, H.: "The effect of gasoline prices on household location"

2010/29, Koethenbuerger, M.: "How do local governments decide on public policy in fiscal federalism? Tax vs. expenditure optimization"

2010/30, Abel, J.; Dey, I.; Gabe, T.: "Productivity and the density of human capital"

2010/31, Gerritse, M.: "Policy competition and agglomeration: a local government view"

2010/32, Hilber, C.; Lyytikäinen, T.; Vermeulen, W.: "Capitalization of central government grants into local house prices: panel data evidence from England"

2010/33, Hilber, C.; Robert-Nicoud, F.: "On the origins of land use regulations: theory and evidence from us metro areas"

2010/34, Picard, P.; Tabuchi, T.: "City with forward and backward linkages"

2010/35, Bodenhorn, H.; Cuberes, D.: "Financial development and city growth: evidence from Northeastern American cities, 1790-1870"



ieb@ub.edu www.ieb.ub.edu

Fiscal Federalism

Document de treball de l'IEB